Author's response to reviews

Title: Eating Patterns of Turkish Adolescents Compared With Recommendations

Authors: mehmet akman (makman4@gmail.com)
hulya akan (hakan@yeditepe.edu.tr)
guldal izbirak (gizbirak@yeditepe.edu.tr)
ozlem tanriover (otanriover@yeditepe.edu.tr)
sirma mine tilev (stilev@yeditepe.edu.tr)
anil yildiz (ayildiz@yeditepe.edu.tr)
simge tektas (stektas@yeditepe.edu.tr)
osman hayran (ohayran@yeditepe.edu.tr)
ayca vitrinel (avitrinel@yeditepe.edu.tr)

Version: 2 Date: 5 February 2010

Author's response to reviews:

Dear Editor,

we have received an email requiring revision in the methods section for ethical approval. This approval is given by the national education directory of istanbul governorship (date 03.04.2009, ref numer:B.08.4.MEM.4.34.00.04.540/39208-397)under condition of taking informed consent from the parents and children. We have added the name of the body to the methods section (last sentence of third paragraph in methods section). If required we can also add the reference number and date of approval.

yours sincerely

mehmet akman